Author's response to reviews

Title: Anaesthesia for serial whole-lung lavage in a patient with severe pulmonary alveolar proteinosis

Authors:

Adrian JR Evans (adrian.evans@doctors.net.uk)
James Varley (varley_james@hotmail.com)
Stephen T Webb (stephentwebb@doctors.org.uk)
Andrew A Klein (andrew.klein@papworth.nhs.uk)

Version: 3 Date: 3 March 2008

Author's response to reviews: see over
14 January 2008

Dear Sir/Madam

Title: Anaesthesia for serial whole-lung lavage in a patient with severe pulmonary alveolar proteinosis

The undersigned:

a) have contributed to the above paper and are familiar with the final draft

b) have no conflict of interest

c) state that the paper has not been published or is for consideration for publication elsewhere;

Patient consent has been sought and received.

Sincerely

Adrian JR Evans

A James Varley

Stephen T Webb

Andrew A Klein

Email for correspondence: stephentwebb@doctors.org.uk